Author’s response to reviews

Title: Densities in the left innominate vein after removal of an implantable venous device: the importance of a differential diagnosis

Authors:

James van Bastelaar (j.van.bastelaar@umcg.nl)
Carolien HC Janssen (c.h.c.janssen@umcg.nl)
Eveline SJM de Bont (e.s.j.m.de.bont@umcg.nl)
Nicole M Blijlevens (n.blijlevens@hemat.umcn.nl)
Robertine van Baren (r.van.baren@umcg.nl)

Version: 2 Date: 1 February 2012

Author’s response to reviews: see over
TO: Professor Michael Kidd AM, Editor-in-chief Journal of Medical Case Reports

FROM: James van Bastelaar, M.D.
Department of Surgery
University Medical Center Groningen
PO BOX 30.001
9700 RB Groningen, the Netherlands

Re. Re-submission to Journal of Medical Case Reports

Groningen, 1 February 2012.

Dear Professor Kidd,

Hereby we would like to submit our manuscript ‘Densities in the left innominate vein after removal of an implantable venous device: the importance of a differential diagnosis’

As requested, the following changes have been made to this manuscript:
- informed consent has been included
- abstract word count, text word count and number of figures in the title page have been removed
- the keywords section after the abstract has been removed
- the Abstract has been restructured into the following three sections: Introduction, Case presentation, and Conclusion.
- ethnicity of the patient has been included in the abstract and case presentation sections of the manuscript.
- figures embedded in the main manuscript have been removed
- a conclusion section has been included as the last section of the text
- competing interests section has been added
- the authors contribution section has been added
- a figure title and legend section has been added after the reference list

We hope you would like to consider the present manuscript for publication in the Journal of Medical Case Reports. Should you have any questions regarding this work, please feel free to contact me at any time.

Yours sincerely,

James van Bastelaar